\documentclass[12pt,reqno]{amsart}

\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{amssymb}
\usepackage{amscd}
%\usepackage{showkeys}

%
% constants
%
\setlength{\textwidth}{160mm}
\setlength{\textheight}{230mm}
\setlength{\oddsidemargin}{0mm}
\setlength{\evensidemargin}{0mm}
\setlength{\topmargin}{0mm}

%
% Theorems
%
\theoremstyle{definition}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{prop}[theorem]{Proposition}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{question}[theorem]{Question}
\newtheorem{problem}[theorem]{Problem}

\numberwithin{equation}{section}
%
% Proof
%
\newenvironment{demo}[1]{%
  \trivlist
  \item[\hskip\labelsep
        {\it #1.}]
}{%
\hfill\qedsymbol
  \endtrivlist
}

%
% Symbols
%
\newcommand\Pf{\operatorname{Pf}}
\newcommand\sgn{\operatorname{sgn}}
\newcommand\trans{{}^t\!}
\newcommand\ep{\varepsilon}
\renewcommand\tilde{\widetilde}
\renewcommand\hat{\widehat}

\allowdisplaybreaks

\begin{document}
\begin{thebibliography}{WW}

\bibitem{Bazin}
M.~Bazin,
Sur une question relative aux d\'eterminants, 
J. Math. Pures Appl. {\bfseries 16} (1851), 145--160.

\bibitem{Giambelli}
G.~Z.~Giambelli,
Alcune propriet\`a delle funzioni simmetriche caratteristiche, 
Atti Accad. Sci. Torino Cl. Sci. Fis. Mat. Natur. {\bfseries 38} (1903), 323--344.

\bibitem{HG}
A.~M. Hamel and I.~P.~Goulden,
Planar decompositions of tableaux and Schur function determinants,
European J. Combin. {\bfseries 16} (1995), 461--477.

\bibitem{Hirota}
R.~Hirota,
Generalizations of determinant identities by Pfaffians,
in ``Mathematical Theories and Applications of Nonlinear Waves and Nonlinear Dynamics'',
Reports of RIAM Symposium No.~15ME-S3, 2004, pp.~148--156 (in Japanese).

\bibitem{IIO}
M.~Ishikawa, M.~Ito and S.~Okada,
A compound determinant identity for rectangular matrices and determinants of Schur functions,
Adv. in Appl. Math. {\bfseries 51} (2013), 635--654.

\bibitem{IsO}
M.~Ishikawa and S.~Okada,
Identities for determinants and Pfaffians, and their applications,
Sugaku Expositions {\bf 27} (2014), 85--116.

\bibitem{ItO}
M.~Ito and S.~Okada,
An application of Cauchy--Sylvester's theorem on compound determinants 
to a $BC_n$-type Jackson integral,
in ``Partitions, $q$-Series, and Modular Forms'', Dev. Math. {\bfseries 23},
Springer, New York, 2012, pp.~145--157.

\bibitem{Jacobi}
C.~G.~J.~Jacobi,
De binis quibuslibet functionibus homogeneis secundi ordinis per
substitutiones lineares in alias binas transformandis, quae solis
quadratis variabilium constant; una cum variis theorematis de
transformatione et determinatione integralium multiplicium,
J. reine angew. Math. {\bfseries 12} (1833), 1--69.
Gesammelte Werke, vol.~3, pp.~191--268.

\bibitem{JP}
T.~J\'ozefiak and P.~Pragacz, 
A determinantal formula for skew $Q$-functions, 
J. London Math. Soc. (2) {\bfseries 43} (1991), 76--90.

\bibitem{Knuth}
D.~E.~Knuth,
Overlapping Pfaffians,
Electron. J. Combin. {\bfseries 3} (no.~2, The Foata Festschrift) (1996), Article~\#R5, 13~pp.

\bibitem{LP}
A.~Lascoux and P.~Pragacz,
\'Equerres et fonctions de Schur,
C. R. Acad. Sci. Paris S\'er. I Math. {\bfseries 299} (1984), 955--958.

\bibitem{Leclerc}
B.~Leclerc,
On identities satisfied by minors of a matrix,
Adv. Math. {\bfseries 100} (1993), 101--132.

\bibitem{Macdonald}
I.~G.~Macdonald,
``Symmetric Functions and Hall Polynomials,'' 2nd edition,
Oxford Univ. Press, 1995.

\bibitem{NOS}
A.~Nakayashiki, S.~Okada and Y.~Shigyo,
On the expansion coefficients of the KP hierarchy,
J. Integrable Syst. {\bfseries 2} (2017), xyx007.
% arXiv:1704.03659.

\bibitem{Schur}
I.~Schur,
\"Uber die Darstellung der symmetrischen und der alternierenden Gruppe durch gebrochene lineare Substitutionen, 
J. reine angew. Math. {\bfseries 139} (1911), 155--250.

\bibitem{Shigyo}
Y.~Shigyo,
On the expansion coefficients of tau-function of the BKP-hierarchy,
J. Phys.~A {\bfseries 49} (2016), 295--201.

\bibitem{Sylvester}
J.~J.~Sylvester, 
On the relation between the minor determinants of linearly equivalent quadratic functions,
Philos. Magazine (4) {\bfseries 1} (1851) 295--305, 415.
Collected Mathematical Papers, vol.~1, pp.~241--250, 251.

\bibitem{You}
Y.~C.~You,
Polynomial solutions of the BKP hierarchy and projective representations of symmetric groups, 
in ``Infinite Dimensional Lie Algebras and Groups'', Adv. Ser. in Math. Phys. {\bfseries 7}, 
World Sci., 1989, pp.~449--466.

\end{thebibliography}


\title[Generalized Sylvester Formulas and Skew Giambelli Identities]{
Generalized Sylvester Formulas \\
and Skew Giambelli Identities
}

\author{
Soichi OKADA
}

\address{
Graduate School of Mathematics, Nagoya University, 
Furo-cho, Chikusa-ku, Nagoya 464-8602, Japan
}

\email{
okada@math.nagoya-u.ac.jp
}

\thanks{
This work was partially supported by 
JSPS Grants-in-Aid for Scientific Research No.~24340003 and No.~15K13425.
}

\date{}

\keywords{%
Sylvester formula, compound determinant, compound Pfaffian, 
skew Schur function, skew $Q$-function.
}

\subjclass[2010]{Primary~15A15; Secondary~05E05.}

\begin{abstract}
We obtain a common generalization of two types of Sylvester formulas for compound determinants 
and its Pfaffian analogue.
As applications, we give generalizations of the Giambelli identity to skew Schur functions 
and the Schur identity to Schur's skew $Q$-functions.
\end{abstract}

\maketitle

\thispagestyle{myheadings}
\font\rms=cmr8 
\font\its=cmti8 
\font\bfs=cmbx8

\markright{\its S\'eminaire Lotharingien de
Combinatoire \bfs 80 \rms (2021), Article~B80e\hfill}
\def\thepage{}


%%%%%%%%%%
% Introduction
%%%%%%%%%%

\section{%
Introduction
}

Determinant and Pfaffian formulas for general matrices 
are a powerful tool in proving relations among special functions 
and in evaluating specific determinants and Pfaffians.
For example, in \cite{ItO}, we applied the Cauchy--Sylvester formula for compound determinants 
to give a transparent proof of the evaluation of the determinant involving 
$BC_n$-type Jackson integrals.
Furthermore, 
in \cite{IIO}, we found a variant of the Cauchy--Sylvester formula 
and obtained product evaluations of determinants of classical group characters.
The aims of this paper are to establish a common generalization of two types of Sylvester formulas 
for compound determinants and its Pfaffian analogue, and to obtain symmetric function identities 
by applying these generalized formulas.

Given a matrix $X = \bigl( x_{i,j} \bigr)$ and sequences $I = (i_1, \dots, i_p)$ of row indices 
and $I' = (i'_1, \dots, i'_q)$ of column indices, let $X \begin{pmatrix} I \\ I' \end{pmatrix}$ 
be the $p \times q$ matrix 
obtained from $X$ by picking up the rows indexed by $I$ and the columns indexed by $I'$, i.e.,
$$
X \begin{pmatrix} I \\ I' \end{pmatrix}
 =
\left( x_{i_\alpha, i'_\beta} \right)_{1 \le \alpha \le p, \, 1 \le \beta \le q}.
$$
For two sequences $I$ and $J$, we denote by $I \sqcup J$ the concatenation of $I$ and $J$, 
and, if $J$ is a subsequence of $I$, then we denote by $I \setminus J$ 
the sequence obtained by removing the entries of $J$ from $I$.
For example, $(1,3,4) \sqcup (2,6) = (1,3,4,2,6)$ and $(1,3,4,2,6) \setminus (3,2) = (1,4,6)$.
Then the Sylvester formula and its dual version (the Jacobi formula) are stated as follows.

\begin{prop}
\label{prop:Sylvester}
Let $X$ be a matrix and let $K$ and $K'$ be sequences of row and column indices of the same length.
\begin{enumerate}
\item[(1)]
{\sc(Sylvester \cite{Sylvester})}
For sequences $I = (i_1, \dots, i_p)$ of row indices and $I' = (i'_1, \dots, i'_p)$ of column indices, 
we have
\begin{equation}
\label{eq:Sylvester1}
\det \left(
 \det X \begin{pmatrix} ( i_\alpha ) \sqcup K \\ ( i'_\beta ) \sqcup K' \end{pmatrix}
 \right)_{1 \le \alpha, \, \beta \le p}
 =
\det X \begin{pmatrix} I \sqcup K \\ I' \sqcup K' \end{pmatrix}
\cdot
\left( \det X \begin{pmatrix} K \\ K' \end{pmatrix} \right)^{p-1}.
\end{equation}
\item[(2)]
{\sc(Jacobi \cite{Jacobi})}
For sequences $J = (j_1, \dots, j_q)$ of row indices and $J' = (j'_1, \dots, j'_q)$ of column indices, 
we have
\begin{equation}
\label{eq:Sylvester2}
\det \left(
 \det X \begin{pmatrix}
  (J \setminus (j_\alpha)) \sqcup K \\
  (J' \setminus (j'_\beta)) \sqcup K'
 \end{pmatrix}
\right)_{1 \le \alpha, \, \beta \le q}
\\
 =
\left( \det X \begin{pmatrix} J \sqcup K \\ J' \sqcup K' \end{pmatrix} \right)^{q-1}
\cdot
\det X \begin{pmatrix} K \\ K' \end{pmatrix}.
\end{equation}
\end{enumerate}
\end{prop}

%First page headline in AmS-LaTeX for S\'eminaire Lotharingien de Combinatoire
%--restoring the headers and pagenumbering
\pagenumbering{arabic}
\addtocounter{page}{1}
\markboth{\SMALL SOICHI OKADA}{\SMALL GENERALIZED SYLVESTER FORMULAS AND
SKEW GIAMBELLI IDENTITIES}


Let $X$, $I$, $J$, $K$, $I'$, $J'$, $K'$ be as in Proposition~\ref{prop:Sylvester}, 
and let $L$ and $L'$ be sequences of row and column indices of the same length.
If the submatrix $X \begin{pmatrix} I \sqcup K \sqcup L \\ I' \sqcup K' \sqcup L' \end{pmatrix}$ 
of $X$ is non-singular with inverse matrix $Y$, 
then we can use Jacobi's complementary minor formula \cite{Jacobi} to 
rewrite \eqref{eq:Sylvester1} in terms of $Y$ as
$$
\det 
\left(
 Y \begin{pmatrix} 
  (I' \setminus (i'_\alpha)) \sqcup L' \\
  (I \setminus (i_\beta)) \sqcup L
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le p}
 =
\left(
 \det Y \begin{pmatrix} I' \sqcup L' \\ I \sqcup L \end{pmatrix}
\right)^{p-1}
\cdot
 \det Y \begin{pmatrix} L' \\ L \end{pmatrix},
$$
which has the same form as \eqref{eq:Sylvester2}.
Similarly, 
if the submatrix $X \begin{pmatrix} J \sqcup K \sqcup L \\ J' \sqcup K' \sqcup L' \end{pmatrix}$ 
is non-singular with inverse matrix $Z$, then \eqref{eq:Sylvester2} can be written as
$$
\det 
\left(
 Z \begin{pmatrix}
  (j'_\alpha) \sqcup L' \\
  (j_\beta) \sqcup L
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le q}
 =
\det Z \begin{pmatrix} J' \sqcup L' \\ J \sqcup L \end{pmatrix}
\cdot
\left(
 \det Z \begin{pmatrix} L' \\ L \end{pmatrix}
\right)^{p-1}.
$$
In this sense, Equations \eqref{eq:Sylvester1} and \eqref{eq:Sylvester2} 
can be regarded dual to each other.

One of the main results of this paper is the following theorem, 
which is a common generalization of \eqref{eq:Sylvester1} and \eqref{eq:Sylvester2}.
In fact, we can recover \eqref{eq:Sylvester1} (respectively \eqref{eq:Sylvester2}) 
by specializing $q=0$ (respectively $p=0$) in \eqref{eq:main1}.

\begin{theorem}
\label{thm:main1}
Let $X$ be a matrix and $I$, $J$ and $K$ (respectively $I'$, $J'$ and $K'$) sequences 
of row (respectively column) indices of length $p$, $q$ and $r$.
We define a $(p+q) \times (p+q)$ matrix 
$\tilde{X} = \bigl( \tilde{x}_{\alpha, \beta} \bigr)_{1 \le \alpha, \, \beta \le p+q}$ by
$$
\begin{cases}
\tilde{x}_{\alpha, \beta}
 =
\det X \begin{pmatrix}
 (i_\alpha) \sqcup J \sqcup K \\
 (i'_\beta) \sqcup J' \sqcup K'
\end{pmatrix}, 
 &\text{if $1 \le \alpha, \beta \le p$,}
\\
\tilde{x}_{\alpha, p+\beta}
 =
\det X \begin{pmatrix}
 (i_\alpha) \sqcup ( J \setminus (j_\beta) ) \sqcup K \\
 J' \sqcup K'
\end{pmatrix},
 &\text{if $1 \le \alpha \le p$ and $1 \le \beta \le q$,}
\\
\tilde{x}_{p+\alpha, \beta}
 =
\det X \begin{pmatrix}
 J \sqcup K \\
 (i'_\beta) \sqcup ( J' \setminus (j'_\alpha) ) \sqcup K'
\end{pmatrix},
 &\text{if $1 \le \alpha \le q$ and $1 \le \beta \le p$,}
\\
\tilde{x}_{p+\alpha, p+\beta}
 =
- \det X \begin{pmatrix}
 ( J \setminus (j_\beta) ) \sqcup K \\
 (J' \setminus (j'_\alpha) ) \sqcup K'
\end{pmatrix},
 &\text{if $1 \le \alpha, \beta \le q$.}
\end{cases}
$$
Then we have
\begin{equation}
\label{eq:main1}
\det \tilde{X}
 =
(-1)^q
\det X \begin{pmatrix} I \sqcup K \\ I' \sqcup K' \end{pmatrix}
\cdot
\left( \det X \begin{pmatrix} J \sqcup K \\ J' \sqcup K' \end{pmatrix} \right)^{p+q-1}.
\end{equation}
\end{theorem}

For example, if $I = I' = (1,2)$, $J = J' = (3,4)$, $K = K' = (5)$, then Equation \eqref{eq:main1} reads
$$
\det \begin{pmatrix}
 \begin{vmatrix} 1,3,4,5 \\ 1,3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 1,3,4,5 \\ 2,3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 1,4,5 \\ 3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 1,3,5 \\ 3,4,5 \end{vmatrix}
\\[13pt]
 \begin{vmatrix} 2,3,4,5 \\ 1,3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 2,3,4,5 \\ 2,3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 2,4,5 \\ 3,4,5 \end{vmatrix}
&
 \begin{vmatrix} 2,3,5 \\ 3,4,5 \end{vmatrix}
\\[13pt]
 \begin{vmatrix} 3,4,5 \\ 1,4,5 \end{vmatrix}
&
 \begin{vmatrix} 3,4,5 \\ 2,4,5 \end{vmatrix}
&
 - \begin{vmatrix} 4,5 \\ 4,5 \end{vmatrix}
&
 - \begin{vmatrix} 3,5 \\ 4,5 \end{vmatrix}
\\[13pt]
 \begin{vmatrix} 3,4,5 \\ 1,3,5 \end{vmatrix}
&
 \begin{vmatrix} 3,4,5 \\ 2,3,5 \end{vmatrix}
&
 - \begin{vmatrix} 4,5 \\ 3,5 \end{vmatrix}
&
 - \begin{vmatrix} 3,5 \\ 3,5 \end{vmatrix}
\end{pmatrix}
 =
(-1)^2
\begin{vmatrix} 1,2,5 \\ 1,2,5 \end{vmatrix}
\cdot
\begin{vmatrix} 3,4,5 \\ 3,4,5 \end{vmatrix}^{2+2-1},
$$
where $\begin{vmatrix} i_1, \dots, i_p \\ i'_1, \dots, i'_p \end{vmatrix}$ stands for 
$\det X \begin{pmatrix} i_1, \dots, i_p \\ i'_1, \dots, i'_p \end{pmatrix}$.

Knuth \cite{Knuth} gave Pfaffian analogues of the Sylvester and Jacobi formulas 
in Proposition~\ref{prop:Sylvester}.
For a skew-symmetric matrix $Y$ and a sequence $I = (i_1, \dots, i_l)$ of row/column indices, 
we write $Y(I)$ for $Y \begin{pmatrix} I \\ I \end{pmatrix}$.

\begin{prop}
\label{prop:Knuth}
Let $Y$ be a skew-symmetric matrix and $K$ be a sequence of row/column indices of even length.
\begin{enumerate}
\item[(1)]
{\sc(Knuth \cite[2.5]{Knuth})}
For a sequence $I = (i_1, \dots, i_l)$ of row/column indices of even length $l$, we have
\begin{equation}
\label{eq:Knuth1}
\Pf
 \Bigl( 
  \Pf Y( ( i_\alpha, i_\beta ) \sqcup K )
 \Bigl)_{1 \le \alpha < \beta \le l}
 =
\Pf Y(I \sqcup K)
\cdot
\bigl( \Pf Y( K ) \bigr)^{l/2-1}.
\end{equation}
\item[(2)]
{\sc(Knuth \cite[(2.7)]{Knuth})}
For a sequence $J = (j_1, \dots, j_m)$ of row/column indices of even length $m$, we have
\begin{equation}
\label{eq:Knuth2}
\Pf
 \Bigl( 
  \Pf Y( (J \setminus ( j_\alpha, j_\beta )) \sqcup K )
 \Bigr)_{1 \le \alpha < \beta \le m}
 =
\bigl( \Pf Y( J \sqcup K ) \bigr)^{m/2-1}
\cdot
\Pf Y( K ).
\end{equation}
\end{enumerate}
\end{prop}

Another main result is the following common generalization of \eqref{eq:Knuth1} and \eqref{eq:Knuth2}.
If $m=0$ (respectively $l=0$), then Equation \eqref{eq:main2} reduces to \eqref{eq:Knuth1} (respectively \eqref{eq:Knuth2}).
Note that Hirota \cite{Hirota} gave the special case $l=m$.

\begin{theorem}
\label{thm:main2}
Let $Y$ be a skew-symmetric matrix.
Let $l$, $m$ and $n$ be nonnegative integers with the same parity,
and let $I$, $J$ and $K$ be sequences of row/column indices of $Y$ of length $l$, $m$ and $n$ respectively.
We define an $(l+m) \times (l+m)$ skew-symmetric matrix $\tilde{Y}
 = \bigl( \tilde{y}_{\alpha, \beta} \bigr)_{1 \le \alpha, \beta \le l+m}$ by
$$
\begin{cases}
\tilde{y}_{\alpha,\beta}
 =
\Pf Y( (i_\alpha,i_\beta) \sqcup J \sqcup K ),
 &\text{if $1 \le \alpha < \beta \le l$,}
\\
\tilde{y}_{\alpha, l+\beta}
 =
\Pf Y( (i_\alpha) \sqcup ( J \setminus (j_\beta)) \sqcup K ),
 &\text{if $ 1 \le \alpha \le l$ and $1 \le \beta \le m$,}
\\
\tilde{y}_{l+\alpha,l+\beta}
 =
\Pf Y( (J \setminus (j_\alpha, j_\beta)) \sqcup K ),
 &\text{if $1 \le \alpha < \beta \le m$.}
\end{cases}
$$
Then we have
\begin{equation}
\label{eq:main2}
\Pf \tilde{Y}
 =
\Pf Y(I \sqcup K)
\cdot
\bigl( \Pf Y(J \sqcup K) \bigr)^{(l+m)/2-1}.
\end{equation}
\end{theorem}

As an application of our generalized Sylvester formula (Theorem~\ref{thm:main1}), 
we give an extension of the Giambelli identity to skew Schur functions, 
which expresses a skew Schur function $s_{\lambda/\mu}$ 
in terms of a determinant of skew Schur functions of the form $\pm s_{\nu/\mu}$.
Similarly, we use the generalized Knuth formula (Theorem~\ref{thm:main2}) to obtain 
a generalization of the Schur identity to Schur's skew $Q$-functions,
which expresses a skew $Q$-function $Q_{\lambda/\mu}$ 
in terms of a Pfaffian of skew $Q$-functions of the form $\pm Q_{\nu/\mu}$.

This article is organized as follows.
In Section~2, we prove Theorem~\ref{thm:main2} by using a Pfaffian analogue of 
the Desnanot--Jacobi identity.
In Section~3, we derive Theorem~\ref{thm:main1} from Theorem~\ref{thm:main2} by 
using a relation between determinants and Pfaffians.
Applications to symmetric function identities are provided in Sections~4 and 5.

%%%%%%%%%%
% Proof of generalized Sylvester identity for Pfaffians
%%%%%%%%%%

\section{%
Proof of Theorem~\ref{thm:main2}
}

Recall a definition and some properties of Pfaffians (see \cite{IsO} for details).
Given an $2m \times 2m$ skew-symmetric matrix $A = \bigl( a_{ij} \bigr)_{1 \le i, j \le 2m}$, 
the Pfaffian of $A$, denoted by $\Pf A$, is defined by
$$
\Pf A
 =
\sum_{\pi \in F_{2m}}
 \sgn(\pi) a_{\pi(1), \pi(2)} a_{\pi(3), \pi(4)} \cdots a_{\pi(2m-1), \pi(2m)},
$$
where $F_{2m}$ is the subset of the symmetric group $S_{2m}$ given by
$$
F_{2m}
 =
\{ \pi \in S_{2m} : \pi(1) < \pi(3) < \dots < \pi(2m-1), \, \pi(2i-1) < \pi(2i) \, (1 \le i \le m) \}.
$$
For example, if $2m=4$, then we have
$$
\Pf A = a_{12} a_{34} - a_{13} a_{24} + a_{14} a_{23}.
$$
For a $2m \times 2m$ skew-symmetric matrix $A$ and a $2m \times 2m$ matrix $T$, we have
$$
\Pf \left( \trans T A T \right) = \det T \cdot \Pf A.
$$
Hence the Pfaffian is alternating in the sense that 
$\Pf \left( a_{\sigma(i), \sigma(j)} \right) 
 = \sgn(\sigma) \Pf \left( a_{i,j} \right)$ for any permutation $\sigma \in S_{2m}$.

One key ingredient of the proof of Theorem~\ref{thm:main2} is the Pfaffian analogue of the Desnanot--Jacobi formula, 
which is a special case of the basic identity in \cite{Knuth}.

\begin{lemma}
\label{lem:Pf-Dodgeson}
(See e.g. \cite[(1.1)]{Knuth}.)
Let $A$ be a skew-symmetric matrix with rows/columns indexed by $( 1, 2, \dots, 2m )$.
For row/column indices $i < j < k < l$, we have
\begin{equation}
\label{eq:Pf-dodgeson}
\Pf A^{i,j} \cdot \Pf A^{k,l} - \Pf A^{i,k} \cdot \Pf A^{j,l} + \Pf A^{i,l} \cdot \Pf A^{j,k}
 =
\Pf A \cdot \Pf A^{i,j,k,l},
\end{equation}
where $A^{i_1, \dots, i_p}$ denotes the skew-symmetric matrix obtained from $A$ 
by removing rows and columns with indices $i_1, \dots, i_p$.
\end{lemma}

First we show the following identities, which are special cases of Theorem~\ref{thm:main2}.

\begin{lemma}
\label{lem:special}
Let $Y$ be a skew-symmetric matrix.
Let $(a, b, c, d)$ and $K$ be sequences of row/column indices.
\begin{enumerate}
\item[(1)]
If the length of $K$ is even, then we have
\begin{multline}
\label{eq:special22}
\Pf Y( (a,b,c,d) \sqcup K ) \cdot \Pf (K)
 - \Pf Y( (a,d) \sqcup K ) \cdot \Pf Y( (b,c) \sqcup K )
\\
 + \Pf Y( (a,c) \sqcup K ) \cdot \Pf Y( (b,d) \sqcup K )
 =
\Pf Y( (a,b) \sqcup K ) \cdot \Pf Y( (c,d) \sqcup K ).
\end{multline}
\item[(2)]
If the length of $K$ is odd, then we have
\begin{multline}
\label{eq:special13}
\Pf Y( ( a,c,d) \sqcup K ) \cdot \Pf Y( (b) \sqcup K)
 - \Pf Y( ( a,b,d) \sqcup K ) \cdot \Pf Y( (c) \sqcup K)
\\
 + \Pf Y( ( a,b,c) \sqcup K ) \cdot \Pf Y( (d) \sqcup K)
 =
\Pf Y( ( b,c,d) \sqcup K ) \cdot \Pf Y( (a) \sqcup K).
\end{multline}
\end{enumerate}
\end{lemma}

\begin{demo}{Proof}
The first identity \eqref{eq:special22} is a restatement of Lemma~\ref{lem:Pf-Dodgeson}.
We shall prove the second identity \eqref{eq:special13}.
Let $e$ be the first entry of $K$ and let $K' = K \setminus (e)$ be the remaining sequence.
If we write $[i_1, \dots, i_r] = \Pf Y( (i_1, \dots, i_r) \sqcup K' )$, 
then the left hand side of \eqref{eq:special13} is written as
$$
[a,c,d,e] \cdot [b,e] - [a,b,d,e] \cdot [c,e] + [a,b,c,e] \cdot [d,e].
$$
By using Lemma~\ref{lem:Pf-Dodgeson}, we have
\begin{align*}
[a,c,d,e] \cdot [\emptyset] &= [a,c] \cdot [d,e] - [a,d] \cdot [c,e] + [a,e] \cdot [c,d],
\\
[a,b,d,e] \cdot [\emptyset] &= [a,b] \cdot [d,e] - [a,d] \cdot [b,e] + [a,e] \cdot [b,d],
\\
[a,b,c,e] \cdot [\emptyset] &= [a,b] \cdot [c,e] - [a,c] \cdot [b,e] + [a,e] \cdot [b,c],
\end{align*}
where $[\emptyset] = \Pf Y(K')$.
Hence the left hand side of \eqref{eq:special13} is equal to
$$
\frac{ 1 }{ [\emptyset] }
\cdot [a,e]
\left(
 [b,c] \cdot [d,e] - [b,d] \cdot [c,e] + [b,e] \cdot [c,d]
\right),
$$
which turns out to be equal to $[a,e] \cdot [b,c,d,e] = \Pf Y ( (a) \sqcup K ) \cdot \Pf Y( (b,c,d) \sqcup K )$ 
again by using Lemma~\ref{lem:Pf-Dodgeson}.
\end{demo}

\begin{demo}{Proof of Theorem~\ref{thm:main2}}
For three sequences $I$, $J$ and $K$, we denote by $Z(I,J,K)$ the $(l+m) \times (l+m)$ skew-symmetric 
matrix given by
$$
Z(I,J,K)
=
\begin{pmatrix}
 \Bigl( 
  \Pf Y( ( i_\alpha, i_\beta ) \sqcup J \sqcup K )
 \Bigr)_{1 \le \alpha < \beta \le l}
&
 \Bigl(
  \Pf Y( ( i_\alpha ) \sqcup ( J \setminus ( j_\beta ) \sqcup K )
 \Bigr)_{\substack{1 \le \alpha \le l \\ 1 \le \beta \le m}}
\\
 &
 \Bigl(
  \Pf Y( (J \setminus ( j_\alpha, j_\beta )) \sqcup K )
 \Bigr)_{1 \le \alpha < \beta \le m}
\end{pmatrix},
$$
where $l$ and $m$ are the lengths of $I$ and $J$ respectively.
We proceed by induction on $l$ and $m$.

First we consider the case where $l \ge 4$.
In this case, we apply Lemma~\ref{lem:Pf-Dodgeson} to the matrix $A = \tilde{Y} = Z(I,J,K)$ 
with $(i,j,k,l) = (1,2,3,4)$.
By using the induction hypothesis on $l$, we have
\begin{align*}
\tilde{Y}^{\alpha, \beta}
 &=
\Pf Z(I \setminus ( i_\alpha, i_\beta ), J, K)
\\
 &=
\Pf Y( (I \setminus ( i_\alpha, i_\beta )) \sqcup K )
\cdot
\bigl ( \Pf Y( J \sqcup K ) \bigr)^{((l-2)+m)/2-1}
\quad(1 \le \alpha < \beta \le 4),
\end{align*}
and
\begin{align*}
\tilde{Y}^{1,2,3,4}
 &=
\Pf Z(I \setminus ( i_1, i_2, i_3, i_4 ), J, K)
\\
 &=
\Pf Y( (I \setminus ( i_1, i_2, i_3, i_4 )) \sqcup K )
\cdot
\bigl( \Pf Y( J \sqcup K ) \bigr)^{((l-4)+m)/2-1}.
\end{align*}
Hence we see that
\begin{align*}
&
\Pf Z(I,J,K)
\\
 &\
=
\frac{ 1 }
     { \Pf Y((I \setminus (i_1, i_2, i_3, i_4)) \sqcup K) }
\left\{
 \begin{array}{l}
  \Pf Y( (I \setminus ( i_1, i_2 )) \sqcup K )
  \cdot
  \Pf Y( (I \setminus ( i_3, i_4 )) \sqcup K )
 \\
  -
  \Pf Y( (I \setminus ( i_1, i_3 )) \sqcup K )
  \cdot
  \Pf Y( (I \setminus ( i_2, i_4 )) \sqcup K )
 \\
 +
  \Pf Y( (I \setminus ( i_1, i_4 )) \sqcup K )
  \cdot
  \Pf Y( (I \setminus ( i_2, i_3 )) \sqcup K )
 \end{array}
\right\}
\\
&\quad\quad
\times
\bigl( \Pf Y(J \sqcup K) \bigr)^{(l+m)/2-1}.
\end{align*}
Again by applying Lemma~\ref{lem:Pf-Dodgeson} to $A = Y(I \sqcup K)$, 
we obtain the desired formula \eqref{eq:main2} for $Z(I,J,K)$.

Next we consider the case where $m \ge 4$.
In this case we apply Lemma~\ref{lem:Pf-Dodgeson} to the matrix $A = \tilde{Y} = Z(I,J,K)$ 
with $(i,j,k,l) = (l+1,l+2,l+3,l+4)$.
We see that
\begin{gather*}
\tilde{Y}^{l+\alpha,l+\beta}
 =
Z(I, J \setminus ( j_\alpha, j_\beta ), ( j_\alpha, j_\beta ) \sqcup K)
\quad\text{if $(\alpha, \beta) = (1,2)$, $(2,3)$ or $(3,4)$},
\\
\tilde{Y}^{l+1,l+2,l+3,l+4}
 =
Z(I, J \setminus ( j_1, j_2, j_3, j_4 ), ( j_1, j_2, j_3, j_4 ) \sqcup K),
\end{gather*}
and
\begin{itemize}
\item
$\tilde{Y}^{l+1,l+3}$ is equal to the matrix 
obtained from $-Z(I, J \setminus ( j_1, j_3 ), ( j_1, j_3 ) \sqcup K)$ 
by multiplying the $(l+1)$st row/column (corresponding to $j_2$) by $-1$,
\item
$\tilde{Y}^{l+2,l+4}$ is equal to the matrix 
obtained from $-Z(I, J \setminus ( j_2, j_4 ), ( j_2, j_4 ) \sqcup K)$ 
by multiplying the $(l+2)$nd row/column (corresponding to $j_3$) by $-1$,
\item
$\tilde{Y}^{l+1,l+4}$ is equal to the matrix 
obtained from $Z(I, J \setminus ( j_1, j_4 ), ( j_1, j_4 ) \sqcup K)$ 
by multiplying the $(l+1)$st and $(l+2)$nd rows/columns (corresponding to $j_2$ and $j_3$) by $-1$.
\end{itemize}
Hence, by using the induction hypothesis on $m$, we have
\begin{align*}
&
Z(I,J,K)
\\
 &\quad
 =
\frac{ 1 }
     { \Pf Y( I \sqcup (j_1, j_2, j_3, j_4) \sqcup K ) }
\left\{
 \begin{array}{l}
  \Pf Y( I \sqcup (j_1, j_2) \sqcup K )
  \cdot
  \Pf Y( I \sqcup (j_3, j_4) \sqcup K )
 \\
  -
  \Pf Y( I \sqcup (j_1, j_3) \sqcup K )
  \cdot
  \Pf Y( I \sqcup (j_2, j_4) \sqcup K )
 \\
  +
  \Pf Y( I \sqcup (j_1, j_4) \sqcup K )
  \cdot
  \Pf Y( I \sqcup (j_2, j_3) \sqcup K )
 \end{array}
\right\}
\\
 &\quad\quad
 \times
 \bigl( \Pf Y( J \sqcup K ) \bigr)^{(l+m)/2-1}.
\end{align*}
Again by using Lemma~\ref{lem:Pf-Dodgeson} to $A = Y( I \sqcup (j_1, j_2, j_3, j_4) \sqcup K )$ 
with $(i,j,k,l) = (l+1,l+2,l+3,l+4)$, we obtain the desired formula \eqref{eq:main2} for $Z(I,J,K)$.

Now it remains to show the cases where
$$
(l,m) = (1,1), \quad (2,2), \quad (1,3), \quad (3,1) \quad\text{and}\quad (3,3).
$$
The case $(l,m) = (1,1)$ is trivial.
The case $(l,m) = (2,2)$ follows from \eqref{eq:special22} and the cases $(l,m) = (1,3)$ and $(3,1)$ follow 
from \eqref{eq:special13} in Lemma~\ref{lem:special}.
Hence it is enough to consider the case $(l,m) = (3,3)$.

Let $l = m = 3$, $I = ( a, b, c )$, $J = ( d, e, f )$, 
and write $\langle i_1, \dots, i_m \rangle = \Pf Y( ( i_1, \dots, i_m ) \sqcup K)$.
Then we have
$$
Z(I,J,K)
=
\begin{pmatrix}
 0 & \langle a,b,d,e,f \rangle & \langle a,c,d,e,f \rangle 
& \langle a,e,f \rangle & \langle a,d,f \rangle & \langle a,d,e \rangle \\
   & 0                         & \langle b,c,d,e,f \rangle 
& \langle b,e,f \rangle & \langle b,d,f \rangle & \langle b,d,e \rangle \\
   &                           & 0 
& \langle c,e,f \rangle & \langle c,d,f \rangle & \langle c,d,e \rangle \\
   &                           & 
& 0                     & \langle f \rangle     & \langle e \rangle \\
   &                           & 
&                       & 0                     & \langle d \rangle \\
   &                           & 
&                       &                       & 0
\end{pmatrix}.
$$
We apply Lemma~\ref{lem:Pf-Dodgeson} to this matrix, say $W$.
It follows from \eqref{eq:special13} that
$$
\Pf W^{1,2}
 =
\langle c \rangle \cdot \langle d,e,f \rangle,
\quad
\Pf W^{1,3}
 =
\langle b \rangle \cdot \langle d,e,f \rangle,
\quad
\Pf W^{2,3}
 =
\langle a \rangle \cdot \langle d,e,f \rangle.
$$
Moreover, it follows from \eqref{eq:special22} that
$$
\Pf W^{3,4}
 =
\langle a,b,d \rangle \cdot \langle d,e,f \rangle,
\quad
\Pf W^{2,4}
 =
\langle a,c,d \rangle \cdot \langle d,e,f \rangle,
\quad
\Pf W^{1,4}
 =
\langle b,c,d \rangle \cdot \langle d,e,f \rangle.
$$
Hence, again by using \eqref{eq:special13}, we see that
\begin{align*}
&
\Pf W^{1,2} \cdot \Pf W^{3,4} - \Pf W^{1,3} \cdot \Pf W^{2,4} + \Pf W^{1,4} \cdot \Pf W^{2,3}
\\
&\quad
 =
\langle d,e,f \rangle^2
 \cdot
\left(
 \langle a \rangle \cdot \langle b,c,d \rangle
 - \langle b \rangle \cdot \langle a,c,d \rangle
 + \langle c \rangle \cdot \langle a,b,d \rangle
\right)
\\
&\quad
 =
\langle d,e,f \rangle^2 \cdot \langle a,b,c \rangle \cdot \langle d \rangle.
\end{align*}
Since $\Pf W^{1,2,3,4} = \langle d \rangle$, we conclude that
$$
\Pf W
 =
\langle a,b,c \rangle \cdot \langle d,e,f \rangle^2
 =
\Pf Y( ( a,b,c ) \sqcup K) \cdot \bigl( \Pf Y( ( d,e,f ) \sqcup K) \bigr)^2.
$$
This completes the proof of Theorem~\ref{thm:main2}.
\end{demo}

%%%%%%%%%%
% Proof of generalized Sylvester identity for determinants
%%%%%%%%%%

\section{%
Proof of Theorem~\ref{thm:main1}
}

In this section we derive Theorem~\ref{thm:main1} from Theorem~\ref{thm:main2}.
Some determinant formulas can be deduced from Pfaffian formulas by using the following fundamental relation 
between determinants and Pfaffians.

\begin{lemma}
(See e.g. \cite{IsO}.)
\label{lem:det=Pf}
Suppose that $m+n$ is even.
If $M$ is an $m \times n$ matrix, then we have
\begin{equation}
\label{eq:det=Pf1}
\Pf \begin{pmatrix}
 O & M \\ -\trans M & O
\end{pmatrix}
 =
\begin{cases}
 (-1)^{\binom{m}{2}} \det M, &\text{if $m=n$,} \\
 0, &\text{if $m \neq n$.}
\end{cases}
\end{equation}
More generally, if $S$ is an $m \times m$ skew-symmetric matrix 
and $M$ is an $m \times n$ matrix, then we have
\begin{equation}
\label{eq:det=Pf2}
\Pf \begin{pmatrix}
 S & M \\ -\trans M & O
\end{pmatrix}
 =
\begin{cases}
 (-1)^{\binom{m}{2}} \det M, &\text{if $m=n$,} \\
 0, &\text{if $m < n$.}
\end{cases}
\end{equation}
\end{lemma}

\begin{demo}{Proof of Theorem~\ref{thm:main1}}
We put
\begin{alignat*}{3}
A &= X \begin{pmatrix} I \\ I' \end{pmatrix},
&\quad
B &= X \begin{pmatrix} I \\ J' \end{pmatrix},
&\quad
P &= X \begin{pmatrix} I \\ K' \end{pmatrix},
\\
C &= X \begin{pmatrix} J \\ I' \end{pmatrix},
&\quad
D &= X \begin{pmatrix} J \\ J' \end{pmatrix},
&\quad
Q &= X \begin{pmatrix} J \\ K' \end{pmatrix},
\\
R &= X \begin{pmatrix} K \\ I' \end{pmatrix},
&\quad
S &= X \begin{pmatrix} K \\ J' \end{pmatrix},
&\quad
T &= X \begin{pmatrix} K \\ K' \end{pmatrix},
\end{alignat*}
that is,
$$
X = \bordermatrix{
  & I' & J' & K' \cr
I & A  & B  & P \cr
J & C  & D  & Q \cr
K & R  & S  & T
},
$$
and consider the following $2(p+q+r) \times 2(p+q+r)$ skew-symmetric matrix $Y$ 
with rows/columns indexed by $I \sqcup I' \sqcup J \sqcup J' \sqcup K \sqcup K'$:
$$
Y = \bordermatrix{
   & I & I' & J & J' & K & K' \cr
I  & O & A & O & B & O & P \cr
I' & -\trans A & O & -\trans C & O & -\trans R & O \cr
J  &  O & C & O & D & O & Q \cr
J' & -\trans B & O & -\trans D & O & -\trans S & O \cr
K  & 0 & R & O & S & O & T \cr
K' & -\trans P & O & -\trans Q & 0 & -\trans T & O
}.
$$
We apply Theorem~\ref{thm:main2} to the sequences 
$\tilde{I} = I \sqcup I'$, $\tilde{J} = J \sqcup J'$ and $\tilde{K} = K \sqcup K'$.

We compute the entries of the matrix $\tilde{Y}$ given in Theorem~\ref{thm:main2}.
By permuting row/\break columns and then by using \eqref{eq:det=Pf1}, 
we see that, for $1 \le \alpha, \beta \le p$, we have 
\begin{align*}
&
\Pf Y( (i_\alpha, i_\beta) \sqcup \tilde{J} \sqcup \tilde{K} )
\\
 &\quad
=
(-1)^{qr}
\Pf \begin{pmatrix}
O_{q+r+2}
 &
X \begin{pmatrix} (i_\alpha, i_\beta) \sqcup J \sqcup K \\ J' \sqcup K' \end{pmatrix}
\\
- \trans \left(
 X \begin{pmatrix} (i_\alpha, i_\beta) \sqcup J \sqcup K \\ J' \sqcup K' \end{pmatrix} 
\right)
 &
O_{q+r}
\end{pmatrix}
\\
 &\quad
=
0,
\\
&
\Pf Y( (i_\alpha, i'_\beta) \sqcup \tilde{J} \sqcup \tilde{K})
\\
 &\quad
=
(-1)^{qr + (q+r)}
\Pf \begin{pmatrix}
 O_{q+r+1}
&
 X \begin{pmatrix} (i_\alpha) \sqcup J \sqcup K \\ (i'_\beta) \sqcup J' \sqcup K' \end{pmatrix}
\\
 - \trans \left(
 X \begin{pmatrix} (i_\alpha) \sqcup J \sqcup K \\ (i'_\beta) \sqcup J' \sqcup K' \end{pmatrix}
 \right)
&
 O_{q+r+1}
\end{pmatrix}
\\
 &\quad
=
(-1)^{qr + (q+r)} \cdot (-1)^{ \binom{q+r+1}{2} }
\det X \begin{pmatrix} (i_\alpha) \sqcup J \sqcup K \\ (i'_\beta) \sqcup J' \sqcup K' \end{pmatrix},
\\
&
\Pf Y( (i'_\alpha, i'_\beta) \sqcup \tilde{J} \sqcup \tilde{K})
\\
 &\quad
=
(-1)^{r^2 + q(q+r)}
\Pf \begin{pmatrix}
 O_{q+r+2}
&
 -\trans \left(
  X \begin{pmatrix} J \sqcup K \\ (i'_\alpha, i'_\beta) \sqcup J' \sqcup K' \end{pmatrix}
 \right)
\\
 X \begin{pmatrix} J \sqcup K \\ (i'_\alpha, i'_\beta) \sqcup J' \sqcup K' \end{pmatrix}
&
 O_{q+r}
\end{pmatrix}
\\
 &\quad
=
0,
\end{align*}
where $O_n$ denotes the $n \times n$ zero matrix.
Similarly, for $1 \le \alpha \le p$ and $1 \le \beta \le q$, we have
\begin{align*}
\Pf Y( (i_\alpha) \sqcup (\tilde{J} \setminus (j_\beta)) \sqcup \tilde{K})
 &=
(-1)^{ qr + \binom{q+r}{2} } 
\det X \begin{pmatrix} (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\ J' \sqcup K' \end{pmatrix},
\\
\Pf Y( (i_\alpha) \sqcup (\tilde{J} \setminus (j'_\beta)) \sqcup \tilde{K})
 &=
0,
\\
\Pf Y( (i'_\alpha) \sqcup (\tilde{J} \setminus (j_\beta)) \sqcup \tilde{K})
 &=
0,
\\
\Pf Y( (i'_\alpha) \sqcup (\tilde{J} \setminus (j'_\beta)) \sqcup \tilde{K})
 &=
(-1)^{q + qr} \cdot (-1)^{\binom{q+r}{2}}
\det X \begin{pmatrix} J \sqcup K \\ (i'_\alpha) \sqcup (J' \setminus (j'_\beta)) \sqcup K' \end{pmatrix},
\end{align*}
and, for $1 \le \alpha, \beta \le q$, 
\begin{align*}
\Pf Y( (\tilde{J} \setminus ( j_\alpha, j_\beta )) \sqcup \tilde{K})
 &=
0,
\\
\Pf Y( (\tilde{J} \setminus (j_\alpha, j'_\beta)) \sqcup \tilde{K})
 &=
(-1)^{(q-1)r} \cdot (-1)^{\binom{q+r-1}{2}}
\det X \begin{pmatrix}
 (J \setminus (j_\alpha)) \sqcup K \\
 (J' \setminus (j'_\beta)) \sqcup K'
\end{pmatrix},
\\
\Pf Y( (\tilde{J} \setminus ( j'_\alpha, j'_\beta )) \sqcup \tilde{K})
 &=
0.
\end{align*}
If we put
\begin{align*}
U
 &= 
\left(
 \det X \begin{pmatrix}
  (i_\alpha) \sqcup J \sqcup K \\
  (i'_\beta) \sqcup J' \sqcup K'
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le p},
\\
V
 &=
\left(
 \det X \begin{pmatrix}
  (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
  J' \sqcup K'
 \end{pmatrix}
\right)_{1 \le \alpha \le p, 1 \le \beta \le q},
\\
V'
 &=
\left(
 \det X \begin{pmatrix}
  J \sqcup K \\
  (i'_\alpha) \sqcup (J' \setminus (j'_\beta)) \sqcup K'
 \end{pmatrix}
\right)_{1 \le \alpha \le p, 1 \le \beta \le q},
\\
W
 &=
\left(
 \det X \begin{pmatrix}
  (J \setminus (j_\alpha)) \sqcup K \\
  (J' \setminus (j'_\beta)) \sqcup K'
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le q},
\end{align*}
then the matrix $\tilde{Y}$ in Theorem~\ref{thm:main2} is given by
$$
\tilde{Y}
 =
\begin{pmatrix}
O              & \ep U                  & \ep V                    & O \\
- \ep \trans U & O                      & O                        & (-1)^q \ep V' \\
- \ep \trans V & O                      & O                        & (-1)^{q-1} \ep W \\
O              & - (-1)^q \ep \trans V' & -(-1)^{q-1} \ep \trans W & O
\end{pmatrix},
$$
where $\ep = (-1)^{qr+\binom{q+r}{2}}$.
By permuting rows/columns and using Lemma~\ref{lem:det=Pf}, we have
\begin{align*}
\Pf \tilde{Y}
 &=
(-1)^{q(p+q)} \cdot (-1)^{\binom{p+q}{2}}
\det \begin{pmatrix}
\ep U                 & \ep V                    \\
-(-1)^q \ep \trans V' & -(-1)^{q-1} \ep \trans W
\end{pmatrix}
\\
 &=
(-1)^{q(p+q) + \binom{p+q}{2} + (p+q)(qr+ \binom{q+r}{2})} \cdot (-1)^{q(q+1)}
\det \begin{pmatrix}
U         & V          \\
\trans V' & - \trans W
\end{pmatrix}.
\end{align*}
Since we have
\begin{align*}
\Pf Y(\tilde{I} \sqcup \tilde{K})
 &=
(-1)^{pr+\binom{p+r}{2}} \det X \begin{pmatrix}I \sqcup K \\ I' \sqcup K' \end{pmatrix},
\\
\Pf Y(\tilde{J} \sqcup \tilde{K})
 &=
(-1)^{qr+ \binom{q+r}{2}} \det X \begin{pmatrix} J \sqcup K \\ J' \sqcup K' \end{pmatrix},
\end{align*}
we obtain
\begin{multline*}
(-1)^{q(p+q) + \binom{p+q}{2} + (p+q) ( qr + \binom{q+r}{2} ) + q(q+1)}
\det \tilde{X}
 \\
 =
(-1)^{pr + \binom{p+r}{2} + (p+q-1) ( qr + \binom{q+r}{2})}
\det X \begin{pmatrix} I \sqcup K \\ I' \sqcup K' \end{pmatrix}
\cdot
\left( \det X \begin{pmatrix} J \sqcup K \\ J' \sqcup K' \end{pmatrix} \right)^{p+q-1}.
\end{multline*}
Now, by noticing the congruence
$$
pq + \binom{p+q}{2} 
+ qr + \binom{q+r}{2}
+ rp + \binom{r+p}{2}
 \equiv
0
\ \bmod 2,
$$
we complete the proof of Theorem~\ref{thm:main1}.
\end{demo}

From Theorem~\ref{thm:main1} we can derive the following formula due to Bazin \cite{Bazin}.

\begin{corollary}[\sc Bazin \cite{Bazin}]
Let $n$ and $p$ be positive integers such that $p \le n$.
If $Z$ is a matrix with columns indexed by $( 1, 2, \dots, n )$, 
and $I = (i_1, \dots, i_p)$, $J = (j_1, \dots, j_p)$, and $K$ are sequences of row indices 
of length $p$, $p$, and $n-p$ respectively, 
then we have
\begin{multline}
\label{eq:Bazin}
\det \left(
 \det Z \begin{pmatrix}
  (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
  (1, 2, \dots, n)
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le p}
\\
 =
(-1)^{p(p-1)/2}
\det Z \begin{pmatrix} I \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
\cdot
\left( \det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix} \right)^{p-1}.
\end{multline}
\end{corollary}

\begin{demo}{Proof}
We apply Theorem~\ref{thm:main1} to the matrix $X = \left( x_{i,j} \right)$ 
with rows indexed by $I \sqcup J \sqcup K$ 
and columns indexed by $(\overline{1}, \dots, \overline{p}, \allowbreak 1, \dots, p, p+1, \dots, n)$, 
whose entries are given by
$$
\begin{cases}
 x_{i,\overline{j}} = z_{i,j}, &\text{if $1 \le j \le p$,} \\
 x_{i,j} = z_{i,j}, &\text{if $1 \le j \le n$.}
\end{cases}
$$
Let $I' = (\overline{1}, \dots, \overline{p})$, $J' = ( 1, \dots, p)$, $K' = (p+1, \dots, n)$.
Then we have
\begin{gather*}
\det X \begin{pmatrix}
 (i_\alpha) \sqcup J \sqcup K \\
 (\overline{i}) \sqcup J' \sqcup K'
\end{pmatrix}
 =
0
\quad(1 \le \alpha \le p, 1 \le i \le p),
\\
\det X \begin{pmatrix}
 (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
 J' \sqcup K'
\end{pmatrix}
 =
\det Z \begin{pmatrix}
 (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
 (1, 2, \dots, n)
\end{pmatrix}
\quad(1 \le \alpha, \beta \le p),
\\
\det X \begin{pmatrix}
 J \sqcup K \\
 (\overline{i}) \sqcup (J' \setminus (j)) \sqcup K'
\end{pmatrix}
 =
\delta_{i,j} (-1)^{i-1} \det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
\quad(1 \le i, j \le p),
\\
\det X \begin{pmatrix} I \sqcup K \\ I' \sqcup K' \end{pmatrix}
 =
\det Z \begin{pmatrix} I \sqcup K \\ (1, 2, \dots, n) \end{pmatrix},
\\
\det X \begin{pmatrix} J \sqcup K \\ J' \sqcup K' \end{pmatrix}
 =
\det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}.
\end{gather*}
Hence we have
\begin{multline*}
\det \begin{pmatrix}
 O_p
&\kern-.8cm
 \left(
  \det Z \begin{pmatrix}
   (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
   (1, 2, \dots, n)
  \end{pmatrix}
 \right)_{1 \le \alpha, \beta \le p}
\\
 \left(
  \delta_{i,j} (-1)^{i-1} \det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
 \right)_{1 \le i, j \le p}
&
 *
\end{pmatrix}
\\
 =
(-1)^p
\det Z \begin{pmatrix} I \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
\cdot
\left(
 \det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
\right)^{2p-1}.
\end{multline*}
By permuting columns in the determinant of the left hand side, 
we see that the left hand side is equal to
\begin{multline*}
(-1)^{p^2}
\det \left(
 \det Z \begin{pmatrix}
  (i_\alpha) \sqcup (J \setminus (j_\beta)) \sqcup K \\
  (1, 2, \dots, n)
 \end{pmatrix}
\right)_{1 \le \alpha, \beta \le p}
\\
\times
(-1)^{\sum_{i=1}^p (i-1)} 
\left(
 \det Z \begin{pmatrix} J \sqcup K \\ (1, 2, \dots, n) \end{pmatrix}
\right)^p.
\end{multline*}
By cancelling the common factor, we obtain the Bazin formula \eqref{eq:Bazin}.
\end{demo}

\begin{remark}
As is shown in \cite{Leclerc}, many determinant identities 
(including \eqref{eq:Sylvester1}, \eqref{eq:Sylvester2} and \eqref{eq:Bazin}) 
are easily derived from Turnbull's identity.
The generalized Sylvester formula \eqref{eq:main1} too is a special case of Turnbull's identity.
For example, if $X$ is a $7 \times 7$ matrix and $I = I' = (1,2,3)$, $J = J' = (4,5)$, $K = K' = (6,7)$, 
then we can use the tableau notation of \cite{Leclerc} to write \eqref{eq:main1} as
$$
\renewcommand{\arraystretch}{1.3}
\begin{array}{|ccccccc|}
\hline
\fbox{$x_1$} & 2 & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & \fbox{$x_2$} & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & \fbox{$x_3$} & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & \fbox{$4$} & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & x_4 & \fbox{$5$} & x_6 & x_7 \\ 
\hline
\end{array}
 =
\begin{array}{|ccccccc|}
\hline
x_1 & x_2 & x_3 & 4 & 5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
1 & 2 & 3 & x_4 & x_5 & x_6 & x_7 \\ 
\hline
\end{array}
$$
where $x_1, \dots, x_7$ are the column vectors of $X$ 
and $1, \dots, 7$ are the column vectors of the $7 \times 7$ identity matrix.
\end{remark}

%%%%%%%%%%
% Applications to symmetric functions
%%%%%%%%%%

\section{%
Skew generalizations of the Giambelli identity
}

In this section, we use the generalized Sylvester formula (Theorem~\ref{thm:main1}) 
to obtain a skew generalization of the Giambelli identity to skew Schur functions.

A partition of a nonnegative integer $n$ is a weakly decreasing sequence 
$\lambda = (\lambda_1, \lambda_2, \dots)$ 
of nonnegative integers such that $|\lambda| = \sum_{i \ge 1} \lambda_i = n$.
The length, denoted by $l(\lambda)$, of a partition $\lambda$ 
is the number of nonzero entries of $\lambda$.
We sometimes write $\lambda = (\lambda_1, \dots, \lambda_{l(\lambda)})$ by omitting the $0$s at the end.
We identify a partition $\lambda$ of $n$ with its Young diagram, 
which is a left-justified array of $n$ cells with $\lambda_i$ cells in the $i$th row.
We denote by $\emptyset$ the empty partition $(0,0,\dots)$ of $0$.
Given a partition $\lambda$, we put
$$
p(\lambda) = \# \{ i : \lambda_i \ge i \},
\quad
\alpha_i = \lambda_i - i,
\quad
\beta_i = \lambda'_i - i
\quad(1 \le i \le p(\lambda)),
$$
where $\lambda'_i$ is the number of cells in the $i$th column of the Young diagram of $\lambda$.
Then we write $\lambda = (\alpha_1, \dots, \alpha_{p(\lambda)} | \beta_1, \dots, \beta_{p(\lambda)})$ 
and call it the Frobenius notation of $\lambda$.

Let $s_\lambda$ be the Schur function corresponding to a partition $\lambda$, 
and $s_{\lambda/\mu}$ the skew Schur function associated to a pair of partitions $\lambda$ and $\mu$.
Note that $s_\lambda = s_{\lambda/\emptyset}$ and $s_{\mu/\mu} = 1$ 
and that $s_{\lambda/\mu} = 0$ unless $\lambda_i \ge \mu_i$ for $i \ge 1$.
(Refer to \cite[Chapter~I]{Macdonald} for details on Schur functions.)
Giambelli \cite{Giambelli} gave a formula which expresses any Schur function 
as a determinant of Schur functions 
of hook shapes $(a|b)$, and Lascoux and Pragacz \cite{LP} gave a generalization of the Giambelli identity 
to skew Schur functions.

\begin{prop}
\label{prop:schur_det}
\begin{enumerate}
\item[(1)]
{\sc(Giambelli \cite{Giambelli})}
For a partition $\lambda = (\alpha_1, \dots, \alpha_p | \beta_1, \dots, \beta_p)$, we have
\begin{equation}
\label{eq:Giambelli}
s_\lambda = \det \left( s_{(\alpha_i | \beta_j)} \right)_{1 \le i, j \le p}.
\end{equation}
\item[(2)]
{\sc(Lascoux, Pragacz \cite{LP})}
If $\lambda = (\alpha_1, \dots, \alpha_p|\beta_1, \dots, \beta_p)$ 
and $\mu = (\gamma_1, \dots, \gamma_q|\delta_1,\break \dots, \delta_q)$ in Frobenius notation, then we have
\begin{equation}
\label{eq:LP}
s_{\lambda/\mu}
 =
(-1)^q
\det \begin{pmatrix}
 \Bigl(
  s_{(\alpha_i|\beta_j)}
 \Bigr)_{1 \le i, j \le p}
&
 \Bigl(
  h_{\alpha_i - \gamma_j}
 \Bigr)_{\substack{1 \le i \le p \\ 1 \le j \le q}}
\\
 \Bigl( 
  e_{\beta_j - \delta_i}
 \Bigr)_{\substack{1 \le i \le q \\ 1 \le j \le p}}
&
 O
\end{pmatrix},
\end{equation}
where $h_k$ (respectively $e_k$) is the $k$th complete (respectively elementary) symmetric function 
and $h_k = e_k = 0$ for $k < 0$.
\end{enumerate}
\end{prop}

We apply Theorem~\ref{thm:main1} to give another generalization of 
the Giambelli identity to skew Schur functions.
In order to state the identity, we extend the definition of skew Schur functions as follows.

\begin{definition}
\label{def:skewSchur}
Given two nonnegative integer sequences $\alpha = (\alpha_1, \dots, \alpha_p)$ 
and $\beta = (\beta_1, \dots, \beta_p)$ of the same length $p$ 
and a partition $\mu$, we define 
$s_{(\alpha|\beta)/\mu}$ by putting
$$
s_{(\alpha|\beta)/\mu}
 =
(-1)^q
\det \begin{pmatrix}
 \Bigl(
  s_{(\alpha_i|\beta_j)}
 \Bigr)_{1 \le i, j \le p}
&
 \Bigl( 
  h_{\alpha_i - \gamma_j}
 \Bigr)_{\substack{1 \le i \le p \\ 1 \le j \le q}}
\\
 \Bigl(
  e_{\beta_j - \delta_i}
 \bigr)_{\substack{1 \le i \le q \\ 1 \le j \le p}}
&
 O
\end{pmatrix},
$$
where $\mu = (\gamma_1, \dots, \gamma_q|\delta_1, \dots, \delta_q)$ in Frobenius notation.
\end{definition}

If the entries of $\alpha$ (or $\beta$) are not distinct, then $s_{(\alpha|\beta)/\mu} = 0$.
Otherwise, if $\sigma$ and $\tau \in S_p$ are permutations such that
$\alpha_{\sigma(1)} > \cdots > \alpha_{\sigma(p)}$ and $\beta_{\tau(1)} > \cdots > \beta_{\tau(p)}$, 
then by \eqref{eq:LP} we have $s_{(\alpha|\beta)/\mu} = \sgn(\sigma \tau) s_{\lambda/\mu}$, 
where $\lambda$ is a partition given by the Frobenius notation 
$\lambda = (\alpha_{\sigma(1)}, \dots, \alpha_{\sigma(p)} | \beta_{\tau(1)}, \dots, \beta_{\tau(p)})$.
Then we have the following skew generalization of the Giambelli identity.

\begin{theorem}
\label{thm:skewGiambelli}
If two partitions $\lambda$ and $\mu$ are represented as 
$\lambda = (\alpha_1, \dots, \alpha_p|\beta_1, \dots, \beta_p)$ and 
$\mu = (\gamma_1, \dots, \gamma_q|\delta_1, \dots, \delta_q)$ in Frobenius notation, 
then we have
\begin{multline}
\label{eq:skewGiambelli}
s_{\lambda/\mu}
\\
 =
(-1)^q
\det
\begin{pmatrix}
 \Bigl(
  s_{(\alpha_i,\gamma_1, \dots, \gamma_q|\beta_j,\delta_1, \dots, \delta_q)/\mu}
 \Bigr)_{1 \le i, j \le p}
&
 \bigl(
  s_{(\alpha_i, \gamma_1, \dots, \widehat{\gamma_j}, \dots, \gamma_q|\delta_1, \dots, \delta_q)/\mu}
 \Bigr)_{\substack{1 \le i \le p \\ 1 \le j \le q}}
\\
 \Bigl(
  s_{(\gamma_1, \dots, \gamma_q|\beta_j, \delta_1, \dots, \widehat{\delta_i}, \dots, \delta_q)/\mu}
 \Bigr)_{\substack{1 \le i \le q \\ 1 \le j \le p}}
&
 O
\end{pmatrix},
\end{multline}
where the symbol $\hat{a}$ means that $a$ is removed from the sequence.
\end{theorem}

If $\mu = \emptyset$, then \eqref{eq:skewGiambelli} reduces to the Giambelli identity \eqref{eq:Giambelli}.
Note that the nonzero entries of the determinant \eqref{eq:skewGiambelli} are of the form 
$\pm s_{\nu/\mu}$ with $\nu/\mu$ a border strip, i.e., a connected skew Young diagram 
containing no $2 \times 2$ block of cells.
Since these entries are skew Schur functions skewed by a fixed partition $\mu$, 
our determinant formula \eqref{eq:skewGiambelli} is not a case 
of the border strip determinants given by 
Hamel--Goulden \cite{HG}.

\begin{demo}{Proof}
Apply Theorem~\ref{thm:main1} to the matrix
$$
X = \begin{pmatrix}
 \Bigl( s_{(\alpha_i|\beta_j)} \Bigr)_{1 \le i, j \le p}
&
 \Bigl( s_{(\alpha_i|\delta_j)} \Bigr)_{\substack{1 \le i \le p \\ 1 \le j \le q}}
&
 \Bigl( h_{\alpha_i-\gamma_j} \Bigr)_{\substack{1 \le i \le p \\ 1 \le j \le q}}
\\
 \Bigl( s_{(\gamma_i|\beta_j)} \Bigr)_{\substack{1 \le i \le q \\ 1 \le j \le p}}
&
 \Bigl( s_{(\gamma_i|\delta_j)} \Bigr)_{1 \le i, j \le q}
&
 \Bigl( h_{\gamma_i-\gamma_j} \Bigr)_{1 \le i, j \le q}
\\
 \Bigl( e_{\beta_j-\delta_i} \Bigr)_{\substack{1 \le i \le q \\ 1 \le j \le p}}
&
 \Bigl( e_{\delta_j-\delta_i} \Bigr)_{1 \le i, j \le q}
&
 O
\end{pmatrix}
$$
with $I = I' = ( 1, \dots, p )$, $J = J' = ( p+1, \dots, p+q )$ and $K = K' = ( p+q+1, \dots, p+2q )$.
Then by definition we have
\begin{gather*}
\det X \begin{pmatrix} ( i ) \sqcup J \sqcup K \\ ( j ) \sqcup J \sqcup K \end{pmatrix}
 =
(-1)^q s_{(\alpha_i,\gamma_1, \dots, \gamma_q|\beta_j,\delta_1, \dots, \delta_q)/\mu},
\\
\det X \begin{pmatrix} ( i ) \sqcup (J \setminus ( p+j )) \sqcup K \\ J \sqcup K \end{pmatrix}
 =
(-1)^q s_{(\alpha_i, \gamma_1, \dots, \hat{\gamma_j}, \dots, \gamma_q|\delta_1, \dots, \delta_q)/\mu},
\\
\det X \begin{pmatrix} J \sqcup K \\ ( i ) \sqcup (J \setminus ( p+j )) \sqcup K \end{pmatrix}
 =
(-1)^q s_{(\gamma_1, \dots, \gamma_q| \beta_i, \delta_1, \dots, \hat{\delta_j}, \dots, \delta_q)/\mu},
\\
\det X \begin{pmatrix} I \sqcup K \\ I \sqcup K \end{pmatrix}
 =
(-1)^q s_{(\alpha|\beta)/\mu}
 =
(-1)^q s_{\lambda/\mu},
\\
\det X \begin{pmatrix} J \sqcup K \\ J \sqcup K \end{pmatrix}
 =
(-1)^q s_{(\gamma|\delta)/\mu}
 =
(-1)^q s_{\mu/\mu}
 =
(-1)^q.
\end{gather*}
Since $X \begin{pmatrix} (J \setminus ( p+i )) \sqcup K \\ (J \setminus ( p+j )) \sqcup K \end{pmatrix}$ 
is a $(2s-1) \times (2s-1)$ matrix whose bottom-right block is the $s \times s$ zero matrix, 
we see that
$$
\det X \begin{pmatrix} (J \setminus ( p+i )) \sqcup K \\ (J \setminus ( p+j )) \sqcup K \end{pmatrix}
 =
0.
$$
Hence, by applying Theorem~\ref{thm:main1}, we obtain the desired identity.
\end{demo}

\begin{remark}
\label{rem:KP}
As is shown in \cite{NOS}, Theorem~\ref{thm:skewGiambelli} is obtained by using 
the Giambelli-type determinant formula for the expansion coefficients 
of the $\tau$-function $\tau(x)$ of the KP hierarchy (\cite[Theorem~1.1]{NOS}) 
and the fact that $\tau(x) = \sum_\lambda s_{\lambda/\mu}(y) s_\lambda(x)$ is a solution of 
the KP hierarchy.
\end{remark}

%%%%%%%%%%
% Applications to symmetric functions
%%%%%%%%%%

\section{%
Skew generalizations of the Schur identity
}

In this section we use Theorem~\ref{thm:main2} to obtain a skew generalization 
of the Schur Pfaffian identity for Schur $Q$-functions.

A partition $\lambda$ is called strict if $\lambda_1 > \lambda_2 > \dots > \lambda_{l(\lambda)}$.
Let $Q_\lambda$ be the Schur $Q$-function corresponding to a strict partition $\lambda$, 
and $Q_{\lambda/\mu}$ the skew $Q$-function associated 
with a pair of strict partitions $\lambda$ and $\mu$.
Note that $Q_{\lambda/\emptyset} = Q_\lambda$, $Q_{\mu/\mu} = 1$ and 
$Q_{\lambda/\mu} = 0$ unless $\lambda_i \ge \mu_i$ for all $i$.
See \cite[Chapter~III, Section~8]{Macdonald} for details on Schur $Q$-functions.

Schur \cite{Schur} defined the Schur $Q$-function corresponding to any strict partition 
as a Pfaffian of Schur $Q$-functions corresponding to strict partitions with at most two rows,
and J\'ozefiak and Pragacz \cite{JP} gave a generalization of Schur's identity to 
skew $Q$-functions.

\begin{prop}
\label{prop:Schur}
For sequences $\alpha = (\alpha_1, \dots, \alpha_l)$ and $\beta = (\beta_1, \dots, \beta_m)$ 
of nonnegative integers, let $S^\alpha_\beta$ and $T^\alpha_\beta$ be 
$l \times m$ matrices defined by
$$
S^\alpha_\beta
 =
\Bigl( Q_{(\alpha_i,\beta_j)} \Bigr)_{1 \le i \le l, 1 \le j \le m},
\quad
T^\alpha_\beta
 =
\Bigl( Q_{(\alpha_i - \beta_{m+1-j})} \Bigr)_{1 \le i \le l, 1 \le j \le m},
$$
where we use the conventions
$$
Q_{(a,b)} = - Q_{(b,a)},
\quad
Q_{(a,0)} = - Q_{(0,a)} = Q_{(a)},
\quad
Q_{(0,0)} = 0
$$
for positive integers $a$ and $b$, and $Q_{(a)} = 0$ for $a < 0$.
Then we have the following Pfaffian identities:
\begin{enumerate}
\item[(1)]
{\sc(Schur \cite{Schur})}
If $\lambda$ is a strict partition, then we have
\begin{equation}
\label{eq:Schur}
Q_\lambda
 = 
\begin{cases}
 \Pf S^\lambda_\lambda, &\text{if $l(\lambda)$ is even,} \\
 \Pf S^{\lambda^0}_{\lambda^0}, &\text{if $l(\lambda)$ is odd,}
\end{cases}
\end{equation}
where $\lambda = (\lambda_1, \dots, \lambda_{l(\lambda)})$ and 
$\lambda^0 = (\lambda_1, \dots, \lambda_{l(\lambda)}, 0)$.
\item[(2)]
{\sc(J\'ozefiak, Pragacz \cite{JP})}
For two strict partitions $\lambda$ and $\mu$, we have
\begin{equation}
\label{eq:JP}
Q_{\lambda/\mu}
 =
\begin{cases}
\Pf \begin{pmatrix}
 S^\lambda_\lambda & T^\lambda_\mu \\
 -\trans T^\lambda_\mu & O
\end{pmatrix},
 &\text{if $l(\lambda) + l(\mu)$ is even,} \\
\Pf \begin{pmatrix}
 S^\lambda_\lambda & T^\lambda_{\mu^0} \\
 -\trans T^\lambda_{\mu^0} & O
\end{pmatrix},
 &\text{if $l(\lambda) + l(\mu)$ is odd.}
\end{cases}
\end{equation}
\end{enumerate}
\end{prop}

We find another generalization of the Schur identity \eqref{eq:Schur} to skew $Q$-functions.
To state the identity, we extend the definition of skew $Q$-functions as follows.

\begin{definition}
\label{def:skewQ}
Given a nonnegative integer sequence $\alpha$ of length $l$ and a strict partition $\mu$, 
we define $Q_{\alpha/\mu}$ by putting
\begin{align*}
Q_{\alpha/\mu}
 &=
\begin{cases}
\Pf \begin{pmatrix}
 S^\alpha_\alpha & T^\alpha_\mu \\
 -\trans T^\alpha_\mu & O
\end{pmatrix},
 &\text{if $l + l(\mu)$ is even,} \\
\Pf \begin{pmatrix}
 S^\alpha_\alpha & T^\alpha_{\mu^0} \\
 -\trans T^\alpha_{\mu^0} & O
\end{pmatrix},
 &\text{if $l + l(\mu)$ is odd.}
\end{cases}
\end{align*}
\end{definition}

We note that, if $l+l(\mu)$ is odd, then we have
$$
\begin{pmatrix}
 S^\alpha_\alpha & T^\alpha_{\mu^0} \\
 -\trans T^\alpha_{\mu^0} & O
\end{pmatrix}
 =
\begin{pmatrix}
 S^{\alpha^0}_{\alpha^0} & T^{\alpha^0}_\mu \\
 -\trans T^{\alpha^0}_\mu & O
\end{pmatrix},
$$
where $\alpha^0 = (\alpha_1, \dots, \alpha_l, 0)$.
If the entries of $\alpha$ are not distinct, then $Q_{\alpha/\mu} = 0$.
Otherwise, if $\sigma \in S_l$ is a permutation such that 
$\alpha_{\sigma(1)} > \cdots > \alpha_{\sigma(l)}$, 
then $Q_{\alpha/\mu} = \sgn(\sigma) Q_{\lambda/\mu}$ 
with $\lambda = (\alpha_{\sigma(1)}, \dots, \alpha_{\sigma(l)})$.
Then we have the following skew-generalization of Schur identity \eqref{eq:Schur}.

\begin{theorem}
\label{thm:skewSchur}
For strict partitions $\lambda$ and $\mu$, we have
\begin{equation}
\label{eq:skewSchur}
Q_{\lambda/\mu}
 =
\Pf \begin{pmatrix}
 \Bigl(
  Q_{(\lambda_i,\lambda_j,\mu_1, \dots, \mu_m)/\mu}
 \Bigr)_{1 \le i, j \le l}
&
 \Bigl(
  Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_s)/\mu}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le s}}
\\
 - \trans \Bigl(
  Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_s)/\mu}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le s}}
&
 O
\end{pmatrix},
\end{equation}
where $l = l(\lambda)$, $m = l(\mu)$ and $s = m$ or $m+1$ according to whether $l+m$ is even or odd.
\end{theorem}

If $\mu = \emptyset$, then \eqref{eq:skewSchur} reduces to the Schur identity \eqref{eq:Schur}.

\begin{demo}{Proof}
We put $l = l(\lambda)$ and $m = l(\mu)$.

First we consider the case where $l \equiv m \bmod 2$.
In this case we apply Theorem~\ref{thm:main2} to the $(l+2m) \times (l+2m)$ skew-symmetric matrix
$$
Y
 =
\begin{pmatrix}
 S^\lambda_\lambda & S^\lambda_\mu & T^\lambda_\mu \\
 -\trans S^\lambda_\mu & S^\mu_\mu & T^\mu_\mu \\
 -\trans T^\lambda_\mu & -\trans T^\mu_\mu & O
\end{pmatrix},
$$
with $I = (1, \dots, l)$, $J = (l+1, \dots, l+m)$ and $K = (l+m+1, \dots, l+2m)$.
Then by definition we have
\begin{gather*}
\Pf Y( ( i, j ) \sqcup J \sqcup K )
 =
Q_{(\lambda_i, \lambda_j, \mu_1, \dots, \mu_m)/\mu},
\\
\Pf Y( ( i ) \sqcup (J \setminus ( l+j )) \sqcup K )
 =
Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)/\mu}.
\end{gather*}
Since $Y( (J \setminus ( l+i, l+j )) \sqcup K)$ is a $(2m-2) \times (2m-2)$ skew-symmetric matrix 
whose bottom-right block is the $m \times m$ zero matrix, we have 
$\Pf Y( (J \setminus ( l+i, l+j )) \sqcup K) = 0$ by \eqref{eq:det=Pf2}.
Also we have
$$
\Pf Y( I \sqcup K) = Q_{\lambda/\mu},
\quad
\Pf Y( J \sqcup K) = Q_{\mu/\mu} = 1.
$$
Hence, by applying \eqref{eq:main2}, we obtain \eqref{eq:skewSchur}.

Next we consider the case where $l \not\equiv m \bmod 2$.
In this case, we apply Theorem~\ref{thm:main2} to the $(l+1+2m) \times (l+1+2m)$ skew-symmetric matrix
$$
Y
 =
\begin{pmatrix}
 S^{\lambda^0}_{\lambda^0} & S^{\lambda^0}_\mu & T^{\lambda^0}_\mu \\
 -\trans S^{\lambda^0}_\mu & S^\mu_\mu & T^\mu_\mu \\
 -\trans T^{\lambda^0}_\mu & -\trans T^\mu_\mu & O
\end{pmatrix},
$$
with $I = (1, \dots, l+1)$, $J = (l+2, \dots, l+m+1)$ and $K = (l+m+2, \dots, l+2m+1)$ 
and $\lambda_{l+1} = 0$.
Then by definition we have
\begin{gather*}
\Pf Y( ( i, j ) \sqcup J \sqcup K )
 =
Q_{(\lambda_i,\lambda_j,\mu_1, \dots, \mu_m)/\mu}
\quad\text{if $1 \le i < j \le l$,}
\\
\Pf Y( (i) \sqcup (J \setminus (l+1+j)) \sqcup K )
 =
Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)/\mu}
\quad\text{if $1 \le i \le l$ and $1 \le j \le m$.}
\end{gather*}
If $1 \le i \le l$, then by moving the $2$nd column/row to the $(m+1)$st column/row we see that
\begin{align*}
\Pf Y( ( i, l+1 ) \sqcup J \sqcup K )
 &=
(-1)^m
\Pf \begin{pmatrix}
 S^{(\lambda_i, \mu_1, \dots, \mu_m)}_{(\lambda_i,\mu_1, \dots, \mu_m)}
&
 T^{(\lambda_i, \mu_1, \dots, \mu_m)}_{(\mu_1, \dots, \mu_m, 0)}
\\
 -\trans T^{(\lambda_i, \mu_1, \dots, \mu_m)}_{(\mu_1, \dots \mu_m,0)}
&
 O
\end{pmatrix}
\\
 &=
(-1)^m Q_{(\lambda_i,\mu_1, \dots, \mu_m)/\mu}.
\end{align*}
If $1 \le j \le m$, then by moving the $1$st row/column to the $m$th row/column 
and by noting $Q_{(\lambda_{l+1}-\mu_k)} = 0$ for $1 \le k \le m$, we have
$$
\Pf Y( ( l+1 ) \sqcup (J \setminus ( l+1+ j )) \sqcup K )
 =
(-1)^{m-1}
\Pf \begin{pmatrix}
 S^{(\mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)}_{(\mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)}
&
 T^{(\mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)}_{(\mu_1, \dots, \mu_m,0)}
\\
 -\trans T^{(\mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)}_{(\mu_1, \dots, \mu_m,0)}
&
 O_{m+1}
\end{pmatrix}.
$$
Hence by using \eqref{eq:det=Pf2}, we have 
$\Pf Y( ( l+1 ) \sqcup (J \setminus ( l+1+ j )) \sqcup K ) = 0$.
Also we have
$$
\Pf Y( I \sqcup K) = Q_{\lambda/\mu},
\quad
\Pf Y( J \sqcup K) = Q_{\mu/\mu} = 1.
$$
Therefore by applying \eqref{eq:main2}, we obtain
$$
Q_{\lambda/\mu} = \Pf \tilde{Y},
$$
where the entries of the skew-symmetric matrix 
$\tilde{Y} = \bigl( \tilde{y}_{i,j} \bigr)_{1 \le i, j \le l+m+1}$ 
are given by
$$
\begin{cases}
\tilde{y}_{i,j} = Q_{(\lambda_i,\lambda_j,\mu_1, \dots, \mu_m)/\mu}, &\text{if $1 \le i, j \le l$,} \\
\tilde{y}_{i,l+1} = (-1)^m Q_{(\lambda_i,\mu_1, \dots, \mu_m)/\mu}, &\text{if $1 \le i \le l$,} \\
\tilde{y}_{i,l+1+j} = Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_m)/\mu},
 &\text{if $1 \le i \le l$ and $1 \le j \le m$,} \\
\tilde{y}_{i,j} = 0, &\text{if $l+1 \le i, j \le l+m+1$.}
\end{cases}
$$
By pulling out the common factor $(-1)^m$ from the $(l+1)$st row/column 
and moving the $(l+1)$st row/column to the last row/column, we see that
\begin{multline*}
\Pf \tilde{Y}
 =
(-1)^m \cdot (-1)^m
\\
\times
\Pf \begin{pmatrix}
 \Bigl(
  Q_{(\lambda_i,\lambda_j,\mu_1, \dots, \mu_m)/\mu}
 \Bigr)_{1 \le i, j \le l}
&
 \Bigl(
  Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_{m+1})/\mu}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le m+1}}
\\
 - \trans \Bigl(
  Q_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_{m+1})/\mu}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le m+1}}
&
 O
\end{pmatrix}.
\end{multline*}
This completes the proof of Theorem~\ref{thm:skewSchur}.
\end{demo}

\begin{remark}
\label{rem:BKP}
(This remark is due to A.~Nakayashiki.) 
Theorem~\ref{thm:skewSchur} can be also obtained from the theory of the BKP hierarchy 
by using the same idea as in \cite{NOS} (see Remark~\ref{rem:KP}).
We fix a strict partition $\mu$ and consider a formal power series of the form
$$
\tau(x)
 =
Q_\mu \left( \frac{x}{2} \right)
 +
\sum_\lambda \xi_\lambda Q_\lambda \left( \frac{x}{2} \right),
$$
where $\lambda$ runs over strict partitions with $|\lambda| > |\mu|$ 
and $x = (x_1, x_3, x_5, \dots)$ is the so-called Sato variables, 
i.e., $x_i = p_i/i$ in the symmetric function language.
Then Shigyo \cite[Theorem~3]{Shigyo} proves that $\tau(x)$ 
is a solution of the BKP hierarchy if and only if
the coefficients $\xi_\lambda$ satisfy the following Pfaffian formulas:
$$
\xi_\lambda
 =
\Pf \begin{pmatrix}
 \Bigl(
  \xi_{(\lambda_i,\lambda_j,\mu_1, \dots, \mu_m)}
 \Bigr)_{1 \le i, j \le l}
&
 \Bigl(
  \xi_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_s)}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le s}}
\\
 - \trans \Bigl(
  \xi_{(\lambda_i, \mu_1, \dots, \hat{\mu_j}, \dots, \mu_s)}
 \Bigr)_{\substack{1 \le i \le l \\ 1 \le j \le s}}
&
 O
\end{pmatrix},
$$
where $l = l(\lambda)$, $m=l(\mu)$, and $s = m$ or $m+1$ according to whether $l+m$ is even or odd, 
and we use the convention that $\xi_{(\alpha_1, \dots, \alpha_p)}$ is alternating 
in $\alpha_1, \dots, \alpha_p$.
By using the fact that $Q_\mu(x/2)$ is a solution of the BKP hierarchy \cite{You}, we can show that 
$$
\tau(x)
 = 
\sum_\lambda Q_{\lambda/\mu}(y) Q_\lambda \left( \frac{x}{2} \right)
 =
Q_\mu \left( \frac{x}{2} \right)
\exp \left( \sum_n n x_n y_n \right),
$$
where $n$ runs over all positive odd integers 
and $y = (y_1, y_3, y_5, \dots)$ is another set of variables, 
is a solution of the BKP hierarchy.
Applying Shigyo's formula to this special solution $\tau(x)$, 
we obtain the generalized Schur identity \eqref{eq:skewSchur}.
\end{remark}

\section*{%
Acknowledgment
}

This work is motivated by questions raised by Prof.~A.~Nakayashiki and Dr.~Y.~Shigyo.
The author thanks them for valuable discussions and comments.
 


\end{document}
